\begin{thebibliography}{10}

\bibitem{babuska76}
{\sc Babu\v{s}ka, I., and Aziz, A.~K.}
\newblock On the {A}ngle {C}ondition in the {F}inite {E}lement {M}ethod.
\newblock {\em SIAM Journal on Numerical Analysis 13}, 2 (Apr. 1976), 214--226.

\bibitem{becker81}
{\sc Becker, E.~B., Carey, G.~F., and Oden, J.~T.}
\newblock {\em Finite {E}lements: {A}n {I}ntroduction}.
\newblock Prentice-Hall, Englewood Cliffs, New Jersey, 1981.

\bibitem{bern92}
{\sc Bern, M., and Eppstein, D.}
\newblock Mesh {G}eneration and {O}ptimal {T}riangulation.
\newblock In {\em Computing in Euclidean Geometry}, D.-Z. Du and F.~Hwang,
  Eds., vol.~1 of {\em Lecture Notes Series on Computing}. World Scientific,
  Singapore, 1992, pp.~23--90.

\bibitem{bielak95}
{\sc Bielak, J., Kallivokas, L.~F., Xu, J., and Monopoli, R.}
\newblock Finite {E}lement {A}bsorbing {B}oundary for the {W}ave {E}quation in
  a {H}alfplane with an {A}pplication to {E}ngineering {S}eismology.
\newblock In {\em Third INRIA-SIAM Wave Propogation Conference\/}
  (Juan-les-Pins, France, Apr. 1995).

\bibitem{borkar88}
{\sc Borkar, S., Cohn, R., Cox, G., Gleason, S., Gross, T., Kung, H.~T., Lam,
  M., Moore, B., Peterson, C., Pieper, J., Rankin, L., Tseng, P., Sutton, J.,
  Urbanski, J., and Webb, J.}
\newblock i{W}arp: {A}n {I}ntegrated {S}olution to {H}igh-{S}peed {P}arallel
  {C}omputing.
\newblock In {\em Supercomputing '88\/} (Kissimmee, Florida, Nov. 1988).

\bibitem{carey}
{\sc Carey, G.~F., and Oden, J.~T.}
\newblock {\em Finite {E}lements: {A} {S}econd {C}ourse}.
\newblock Prentice-Hall, Englewood Cliffs, New Jersey, 1983.

\bibitem{chan94}
{\sc Chan, T.~F., and Mathew, T.~P.}
\newblock Domain {D}ecomposition {A}lgorithms.
\newblock In {\em Acta Numerica 1994}. Cambridge University Press, New York,
  1994, pp.~61--143.

\bibitem{clarkson93}
{\sc Clarkson, K.~L., Eppstein, D., Miller, G.~L., Sturtivant, C., and Teng,
  S.-H.}
\newblock Approximating {C}enter {P}oints with {I}terated {R}adon {P}oints.
\newblock In {\em Proceedings of the Ninth Annual Symposium on Computational
  Geometry\/} (San Diego, California, May 1993), Association for Computing
  Machinery, pp.~91--98.

\bibitem{feldmann93}
{\sc Feldmann, A., Stricker, T.~M., and Warfel, T.~E.}
\newblock Supporting {S}ets of {A}rbitrary {C}onnections on i{W}arp through
  {C}ommunication {C}ontext {S}witches.
\newblock In {\em Fifth Annual ACM Symposium on Parallel Algorithms and
  Architectures\/} (Velen, Germany, July 1993), Association for Computing
  Machinery, pp.~203--212.

\bibitem{gremban94}
{\sc Gremban, K.~D., Miller, G.~L., and Teng, S.-H.}
\newblock Moments of {I}nertia and {G}raph {S}eparators.
\newblock In {\em Proceedings of the Fifth Annual ACM-SIAM Symposium on
  Discrete Algorithms\/} (Arlington, Virginia, Jan. 1994), Association for
  Computing Machinery, pp.~452--461.

\bibitem{heath91}
{\sc Heath, M.~T., Ng, E., and Peyton, B.~W.}
\newblock Parallel {A}lgorithms for {S}parse {L}inear {S}ystems.
\newblock {\em SIAM Review 33}, 3 (Sept. 1991), 420--460.

\bibitem{leiserson92}
{\sc Leiserson, C.~E., Abuhamdeh, Z.~S., Douglas, D.~C., Feynman, C.~R.,
  Ganmukhi, M.~N., Hill, J.~V., Hillis, W.~D., Kuszmaul, B.~C., {St. Pierre},
  M.~A., Wells, D.~S., Wong, M.~C., Yang, S.-W., and Zak, R.}
\newblock The {N}etwork {A}rchitecture of the {C}onnection {M}achine {CM-5}.
\newblock In {\em Proceedings of the Fourth Annual ACM Symposium on Parallel
  Algorithms and Architectures\/} (San Diego, California, June 1992),
  Association for Computing Machinery.

\bibitem{lengauer90}
{\sc Lengauer, T.}
\newblock {\em Combinatorial {A}lgorithms for {C}ircuit {L}ayout}.
\newblock Applicable Theory in Computer Science. John Wiley \& Sons,
  Chichester, England, 1990.

\bibitem{mttv93}
{\sc Miller, G.~L., Teng, S.-H., Thurston, W., and Vavasis, S.~A.}
\newblock Automatic {M}esh {P}artitioning.
\newblock In {\em Graph Theory and Sparse Matrix Computation}, A.~George, J.~R.
  Gilbert, and J.~W.~H. Liu, Eds. Springer-Verlag, New York, 1993.

\bibitem{pothen90}
{\sc Pothen, A., Simon, H.~D., and Liou, K.-P.}
\newblock Partitioning {S}parse {M}atrices with {E}igenvectors of {G}raphs.
\newblock {\em SIAM Journal on Matrix Analysis and Applications 11}, 3 (July
  1990), 430--452.

\bibitem{ruppert95}
{\sc Ruppert, J.}
\newblock A {D}elaunay {R}efinement {A}lgorithm for {Q}uality 2-{D}imensional
  {M}esh {G}eneration.
\newblock {\em Journal of Algorithms 18}, 3 (May 1995), 548--585.

\bibitem{shewchuk94}
{\sc Shewchuk, J.~R.}
\newblock An {I}ntroduction to the {C}onjugate {G}radient {M}ethod {W}ithout
  the {A}gonizing {P}ain.
\newblock Available by anonymous FTP to WARP.CS.CMU.EDU (128.2.209.103) as
  quake-papers/painless-conjugate-gradient.ps, Aug. 1994.

\bibitem{shewchuk96b}
{\sc Shewchuk, J.~R.}
\newblock Triangle: {E}ngineering a {2D} {Q}uality {M}esh {G}enerator and
  {D}elaunay {T}riangulator.
\newblock In {\em First Workshop on Applied Computational Geometry\/} (May
  1996), Association for Computing Machinery.

\bibitem{shewchuk93}
{\sc Shewchuk, J.~R., and Ghattas, O.}
\newblock A {C}ompiler for {P}arallel {F}inite {E}lement {M}ethods with
  {D}omain-{D}ecomposed {U}nstructured {M}eshes.
\newblock In {\em Proceedings of the Seventh International Conference on Domain
  Decomposition Methods in Scientific and Engineering Computing\/} (Oct. 1993),
  D.~E. Keyes and J.~Xu, Eds., vol.~180 of {\em Contemporary Mathematics},
  American Mathematical Society, pp.~445--450.

\bibitem{strang73}
{\sc Strang, G., and Fix, G.~J.}
\newblock {\em An {A}nalysis of the {F}inite {E}lement {M}ethod}.
\newblock Prentice-Hall Series in Automatic Computation. Prentice-Hall,
  Englewood Cliffs, New Jersey, 1973.

\bibitem{tmc91}
{\sc {Thinking Machines Corporation}}.
\newblock {\em The {C}onnection {M}achine {CM-5} {T}echnical {S}ummary}.
\newblock 1991.

\end{thebibliography}

\section*{Mesh Generation}

High-quality unstructured meshes are difficult to generate, but not because
it is difficult to generate an unstructured mesh.
There are simple techniques for
triangulating an arbitrary two-dimensional region, and although
tetrahedralizing a three-dimensional region is much more complicated, there
are straightforward techniques for doing so.  A good survey of mesh
triangulation methods in two and three dimensions is provided by
Bern and Eppstein~\cite{bern92}.

The real difficulty, though, is that a mesh generator should avoid producing
elements with large aspect ratios --- ``skinny'' triangles and tetrahedra,
so to speak --- because angles close to $180^\circ$ cause a large
discretization error~\cite{babuska76}, and very small angles cause the
stiffness matrix ${\bf K}$ to be ill-conditioned~\cite{carey}.
This constraint is difficult to maintain in a mesh that must conform to
an object with a complicated shape.

Archimedes includes a two-dimensional mesh generator named
Triangle~\cite{shewchuk96b} that triangulates
complex geometries with high-quality elements, and furthermore allows the
density of elements to vary and be controlled by the user, as was necessary
to generate Figure~\ref{unstruc}.
The meshing algorithm is due to Ruppert~\cite{ruppert95}, who
has proven that his algorithm generates meshes
whose triangles have angles bounded between
$20^\circ$ and $140^\circ$
(except where there are input angles smaller than $20^\circ$; these cannot be
improved).
Archimedes currently has
only rudimentary 3D mesh generation; work on an effective 3D mesh
generator is in progress.

\begin{figure}
\centerline{\epsfxsize=4in \epsfbox{guitar2.2.ele.eps}}
\caption{\label{refine} Refinement of the original mesh.}
\end{figure}

The mesh generation algorithm serves another purpose.
After the mesh is generated and the simulation completed, the
approximate solution might not be
as close to the exact solution as desired.  A better solution
can be found by refining the mesh and repeating the solution process.
A mesh is {\em refined} by dividing some or all of its elements
into a larger number
of smaller elements, as in Figure \ref{refine}.  As the mesh is refined,
the approximate solution
converges to the exact solution $u$.
Numerical analysts have developed
{\em a priori error estimates} that show that the {\em discretization error}
is reduced as the size of the largest edge is reduced.
However, as we have explained, a uniform density of elements
throughout the domain is not always desirable.
Hence, there are also {\em a posteriori error estimates}, which estimate the
discretization error on each element
after the simulation, thereby providing a guide to refinement.
Figure~\ref{refine} was generated from Figure \ref{mesh}
as follows:
Archimedes solved a heat conduction problem on the original mesh,
and generated {\em a posteriori} error estimates on each
element of the mesh.  Based on the error estimates, it determined a maximum
desired triangle area on each element.  Archimedes' mesh generator refined
the existing mesh so that no triangle has area greater
than the maximum allowed.
